Author's response to reviews

Title: Comparison of three different dressings for partial thickness burns in children: study protocol for a randomised controlled trial

Authors:

Emma Gee Kee (e.geekee@uq.edu.au)
Roy M Kimble (royk@uq.edu.au)
Leila Cuttle (l.cuttle@uq.edu.au)
Kellie Stockton (k.stockton@uq.edu.au)

Version: 3 Date: 14 November 2013

Author's response to reviews: see over
Dear Editors-in-Chief,

Thank you for your review of our manuscript entitled “Comparison of three different dressings for partial thickness burns in children: study protocol for a randomised controlled trial” by Emma Gee Kee et al. The required formatting revisions to the manuscript have been made in the form of track changes. Please see the authors’ responses to the reviewer request below:

Question 1: Please state clearly whether or not you have funding in the acknowledgement section. If there is no funding, please state this.

Response: This clinical trial is partially financially supported by a grant given to the University of Queensland by Mölnlycke Healthcare (see page 18 of manuscript).

N.B. Additionally, errors were observed in the Competing Interests section (page 18) of the manuscript by the authors and have been corrected and highlighted with track changes. Therefore, please use the updated manuscript attached if accepted for publication.

I look forward to hearing from you in the near future.

Yours sincerely,

Emma Gee Kee